Author's response to reviews

Title: Enhancement of hypoxia-activated prodrug TH-302 anti-tumor activity by Chk1 inhibition

Authors:

Fanying Meng (fmeng@thresholdpharm.com)
Deepthi Bhupathi (dbhupathi@thresholdpharm.com)
Jessica Sun (jsun@thresholdpharm.com)
Jenny Liu (jliu@thresholdpharm.com)
Dharmendra Ahluwalia (dahluwalia@thresholdpharm.com)
Yan Wang (wyan@thresholdpharm.com)
Mark D Matteucci (MMatteucci@thresholdpharm.com)
Charles P Hart (chart@thresholdpharm.com)

Version: 2 Date: 19 December 2014

Author's response to reviews:

1) Add Line numbers
2) Page numbers were already included in the original submitted version
3) Provided an acknowledgment section